%% Math 131, Fall, 2003
\documentclass[12pt]{article}
\usepackage{latexsym,amsmath,epic,eepic,pstcol}
\usepackage{math131}

\pagestyle{empty}

\begin{document}

\noindent\textbf{Name}:\ \vbox{\hrule width 3in}

\begin{center}
\textbf{Holy Cross College, Fall Semester, 2003} \\
\textbf{Math 131, Quiz \#7, Friday, October 24} \\
Professor Hwang \\
\end{center}

\be
\item Find $\ds\frac{d}{dx} e^x\sin x$
\vfil

\hrulefill

\item Find $\ds\frac{d}{dt}(t^3-3t)^{24}$
\vfil

\hrulefill
\item Let $x$~be a function of~$t$, and suppose
  $\ds\frac{dx}{dt}=\sqrt{x^2+1}$. Compute $\ds\frac{d^2x}{dt^2}$~as a
  function of~$x$.

\vfil\vfil

\ee
\eject

\end{document}
